Author's response to reviews

Title: Cross-talk between alpha1D-adrenoceptors and transient receptor potential vanilloid type 1 triggers prostate cancer cell proliferation.

Authors:

Maria Beatrice Morelli (betatrismorelli@hotmail.com)
Consuelo Amantini (consuelo.amantini@unicam.it)
Massimo Nabissi (massimo.nabissi@unicam.it)
Sonia Liberati (sonia.liberati@studenti.unicam.it)
Claudio Cardinali (claudio.cardinali@uniroma1.it)
Valerio Farfariello (valerio.farfariello@inserm.fr)
Daniele Tomassoni (daniele.tomassoni@unicam.it)
Wilma Quaglia (wilma.quaglia@unicam.it)
Alessandro Piergentili (alessandro.piergentili@unicam.it)
Alessandro Bonifazi (alessandro.bonifazi@unicam.it)
Fabio Del Bello (fabio.delbello@unicam.it)
Matteo Santoni (mattymo@alice.it)
Gabriele Mammana (gabriele.mammana@sanita.marche.it)
Lucilla Servi (lucilla.servi@yahoo.it)
Alessandra Filosa (alessandrafilosa@yahoo.com)
Angela Gismondi (angela.gismondi@uniroma1.it)
Giorgio Santoni (giorgio.santoni@unicam.it)

Version: 5 Date: 18 November 2014

Author's response to reviews: see over
Dear Editors,

we wish to resubmit the revised version of our manuscript MS: 6916364961292275 entitled “Cross-talk between alpha1D-adrenoceptors and transient receptor potential vanilloid type 1 triggers prostate cancer cell proliferation” by Maria Beatrice Morelli, Consuelo Amantini, Massimo Nabissi, Sonia Liberati, Claudio Cardinali, Valerio Farfariello, Daniele Tomassoni, Wilma Quaglia, Alessandro Piergentili, Alessandro Bonifazi, Fabio Del Bello, Matteo Santoni, Gabriele Mammana, Lucilla Servi, Alessandra Filosa, Angela Gismondi and Giorgio Santoni in BMC Cancer.

We answered to the Editor's request and added the additional information required.

We hope that the revised version of our manuscript can be now acceptable for publication in BMC Cancer.

With best regards,

Consuelo Amantini
School of Bioscience and Veterinary Medicine
University of Camerino
via Madonna delle Carceri 9
62032 Camerino (MC) – ITALY
Phone: +390737403312
RESPONSE TO EDITOR'S REQUEST

Editor's Request: "You state patients gave their informed written consent to be included in the prostate cancer database of the Pathology Unit, ASUR 9, however please clarify whether this informed consent covered use of their tissues for research purposes and if so, please include a statement to this effect in your methods."

Answer: We have included the Editor's request in the manuscript. Methods have been accordingly modified (pag. 7, line 170).